Author’s response to reviews

Title: Structural brain changes in First Episode Schizophrenia compared with Fronto-Temporal Lobar Degeneration: a meta-analysis

Authors:

Bayanne Olabi (b.olabi@sms.ed.ac.uk)
Ian Ellison-Wright (ian.ew@inbox.com)
Ed Bullmore (etb23@cam.ac.uk)
Stephen M Lawrie (s.lawrie@ed.ac.uk)

Version: 2 Date: 3 February 2012

Author’s response to reviews:

The manuscript was revised according to PRISMA guidelines, which included adding a flow diagram of the study selection process and the reasons for exclusion at each stage (which is an extra figure in the manuscript).

The ‘Competing interests’ section was placed after the Conclusions.

The Authors’ Contributions section was placed after Competing interests.

An Acknowledgements section was added.